Author’s response to reviews

Title: IgA nephropathy featuring massive wire loop-like deposits in two patients with alcoholic cirrhosis; Case Report

Authors:
Daisuke Takada (daisuke.geotian@gmail.com)
Akinari Sekine (akinari-s@toranomon.gr.jp)
Hiroki Mizuno (hilomiz@yahoo.co.jp)
Junko Nitta (junk.nitta@gmail.com)
Ryo Hazue (mechaike@yahoo.co.jp)
Koki Mise (kokimise@yahoo.co.jp)
Toshiharu Ueno (ueno.toshiharu1978@gmail.com)
Keiichi Sumida (k_sumida0504@yahoo.co.jp)
Masayuki Yamanouchi (yamanouchi.masayuki@gmail.com)
Tatsuya Suwabe (suwabe@toranomon.gr.jp)
Noriko Hayami (non_love_99@yahoo.co.jp)
Junichi Hoshino (hoshino-ind@umin.ac.jp)
Naoki Sawa (naokis@toranomon.gr.jp)
Kenmei Takaichi (takaichi@toranomon.gr.jp)
Takeshi Fujii (tkshfj@gmail.com)
Kenichi Ohashi (ohashi@yokohama-cu.ac.jp)
Yoshifumi Ubara (ubara@toranomon.gr.jp)

Version: 3 Date: 24 Jul 2017

Author’s response to reviews:

Response to editor

1. ok

2. This manuscript was revised by native English corrector. The changes in the text was underlined.
3. According to the editor’s suggestion, multi-panel figures were changed to single-panel figures.
4. Scale bar was inserted to EM figure 1c and 1f, and EM figure 2c and 2f as two arrow heads showing 300 nm in width.